Author's response to reviews

Title: Carnitine deficiency presenting with decreased mental status in a patient with amyotrophic lateral sclerosis receiving long-term tube feeding: a case report

Authors:

Naohi Isse (issenaohi@yahoo.co.jp)
Yoh Miura (miurayou@eagle.ocn.ne.jp)
Toshiyuki Obata (obata@amh ako.hyogo.jp)
Noriko Takahara (norikot@amh ako.hyogo.jp)

Version: 2 Date: 20 April 2013

Author's response to reviews: see over
Professor Michael Kidd  
Editor-in-Chief  
*Journal of Medical Case Reports*

20 April 2013

Dear Professor Kidd,

Re: Manuscript reference No. 4324026378520121

Please find attached a revised version of our manuscript entitled “Carnitine deficiency presenting with decreased mental status in a patient with amyotrophic lateral sclerosis receiving long-term tube feeding: a case report,” which we would like to resubmit for publication as a case report in the *Journal of Medical Case Reports*.

Your comments and those of the reviewers were highly insightful and enabled us to greatly improve the quality of our manuscript.

Revisions in the text are shown using yellow highlight for additions. In accordance with reviewer Ricardo Cruciani’s suggestion, the entire manuscript has been edited by a native English-speaking editor. We hope that the revisions in the manuscript and our accompanying responses will be sufficient to make our manuscript suitable for publication in the *Journal of Medical Case Reports*.

We look forward to hearing from you at your earliest convenience.

Yours sincerely,

Naohi Isse  
Department of Internal Medicine  
Ako City Hospital  
1090 Nakahiro, Ako  
Hyogo 678-0232, Japan  
Tel: +81 791 433222  
Fax: +81 791 430351  
Email: issenaohi@yahoo.co.jp